Author's response to reviews

Title: Infrared micro-spectral imaging: distinction of tissue types in axillary lymph node histology

Authors:

Benjamin Bird (benjamin_bird@hotmail.com)
Milos Miljkovic (milos.miljkovic@yahoo.com)
Melissa J Romeo (m.romeo@neu.edu)
Jennifer Smith (j.smith@medical-research-centre.com)
Nicholas Stone (n.stone@medical-research-centre.com)
Michael W George (mike.george@nottingham.ac.uk)
Max Diem (m.diem@neu.edu)

Version: 10 Date: 14 August 2008

Author's response to reviews:

Changes to affiliation structure, abstract title and end note reference have been made

regards

Dr Benjamin Bird